Author's response to reviews

Title: Scrotal extratesticular schwannoma: a case report and review of the literature

Authors:

Giovanni Palleschi (giovanni.palleschi@uniroma1.it)
Antonio Carbone (antonio.carbone@uniroma1.it)
Jessica Cacciotti (cacciotti.jessica@libero.it)
Giorgia Manfredonia (giorgia.manfredonia@gmail.com)
Natale Porta (linopo@libero.it)
Cosimo De Nunzio (cosimodenunzio@virgilio.it)
Vincenzo Petrozza (vincenzo.petrozza@uniroma1.it)
Antonio Luigi Pastore (antonioluigi.pastore@uniroma1.it)

Version: 2
Date: 1 December 2013

Author's response to reviews:

Dear Editor,

it's a honour for us to submit to Your attention this manuscript which describes a recent observation of a scrotal, extra-testicular Schwannoma that we successfully treated.

We reviewed the literature about this topic and we have noticed that, even if some manuscripts have been already reported with similar cases, there is a lack of a huge discussion which allows to compare them thus to provide a complete evidence of the whole knowledge on this pathologic condition, suggesting the correct clinical and therapeutical approach.

Best regards,

The authors